Author's response to reviews

Title: Recommended practices for computerized clinical decision support and knowledge management in community settings: A qualitative study

Authors:

Joan S Ash (ash@ohsu.edu)
Dean F Sittig (dean.f.sittig@uth.tmc.edu)
Kenneth P Guappone (kenneth.guappone@providence.org)
Richard H Dykstra (ash@ohsu.edu)
Joshua Richardson (jor2032@med.cornell.edu)
Adam Wright (awright5@partners.org)
James Carpenter (james.carpenter@providence.org)
Carmit McMullen (carmit.mcmullen@kpchr.org)
Michael Shapiro (shapirom@ohsu.edu)
Arwen Bunce (buncea@ohsu.edu)
Blackford Middleton (bmiddleton1@partners.org)

Version: 6 Date: 6 February 2012

Author's response to reviews: see over
February 6, 2012

Dear Editor:

We have edited our manuscript titled “Recommended Practices for Clinical Decision Support and Knowledge Management in Community Settings: A Qualitative Study” to comply with your most recent formatting requests.

Formatting changes to be made:

Major revisions (we require the author to make these changes)

Tables: all tables must be created using a Tables tool, and should not be linked files, figures, or generated with tabbed text. Tables must be divided into rows and columns and be oriented in a portrait page layout. Please ensure that text alignment in columns is clear and consistent. We recommend the MS Word auto-format template: simple 1. Please do not include vertical lines in your tables. Please note that we are unable to correctly display merged cells where the merged cell crosses rows: please re-layout your table without these merged elements

We have made these changes.

Minor revisions (we can make these changes for you, although it will speed up publication of your manuscript if you do them while making the major changes above)

Highlighting/tracking: Please remove all highlighting and tracking from the manuscript, and change any coloured text to black

We have done this.

References: please provide the full citation details for references 10 and 26. You can view the BMC reference guide at the following link http://www.biomedcentral.com/bmcgenet/authors/instructions/researcharticle#formatting-references

We have done this.
Please place Figure Legends before the Tables

We have done this.

Figures: It is important for the final layout of the manuscript that the figures are cropped as closely as possible to minimise white space around the image. Our online figure guide contains full details for preparing files for submission and can be viewed here: http://www.biomedcentral.com/ifora/figures

I believe they are cropped as much as possible. Please crop them further if you think it is possible.

Typography: Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions. This is the final proofing stage for your manuscript, and you will not be able to make any changes after acceptance.

We have done this as well.

I hope the paper is ready now.

Sincerely,

Joan S. Ash, Ph.D., M.B.A., M.L.S.

Professor and Vice-Chair